CS 674 – STUDENT INFORMATION

Name:

College and major (undergrads), or field of interest (MEng’s and PhD’s):

Address, phone number, email address:

Relevant coursework (computer science, linguistics, psychology, philosophy):

Do you have access to a computer account you could use for a substantial programming project?

Do you have access to a computer account from which you can download and print a substantial number of papers from the web? (This is in regards to the on-line readings)

What languages are you fluent in?